Author's response to reviews

Title: Clinically relevant safety issues associated with St. John’s wort product labels

Authors:

Kevin A Clauson (clauson@nsu.nova.edu)
Marile L Santamarina (marile_santamarina@pba.edu)
Jennifer C Rutledge (jrutledg@nsu.nova.edu)

Version: 13 Date: 15 July 2008

Author's response to reviews:

Very minor changes were made. The most substantial was that the word counseling was spelled as both counseling and counselling within the paper. This was corrected to one consistent spelling of counseling.